Tutorial for week 5 (Feb 6, 2002)

February 2, 2002

The following questions will be considered during the tutorial sessions on Wednesday...

TAQ1: How can you create a cascade menu item where the cacaded items are radiobuttons (i.e. - only can select one at a time).

TAQ2: Tcl only supports a simple idea of variable scope. How can you create global variables? How can you create local variables? How can you refer to global variables within a proc?

TAQ3: How would you create structured type variables in Tcl? (i.e. a variable with various sub-components). TAQ4: How do you change the name on the surrounding window decoration for an application? TAQ5: How can you create a file-selection dialog box when you click on an ’open’ menu item? How does your appliaction know which file was selected? How does your application know when no file is selected?

TAQ6: How can you change the cursor (to -say- a watch), when it moves over a particular item embedded in a canvas?